Author’s response to reviews

Title: How can we improve priority setting for investments in health research? A case study of tuberculosis

Authors:

Mishal Khan (mishal.khan@lshtm.ac.uk)
Afifah Rahman-Shepherd (afifahrs@gmail.com)
Hannah Painter (Hannah.Painter1@lshtm.ac.uk)
Helen Fletcher (Helen.Fletcher@lshtm.ac.uk)

Version: 2 Date: 20 Jun 2019

Author’s response to reviews:

Dear editor,

We appreciate the constrictive feedback provided by the reviewers and are pleased to confirm that we have been able to address all of the comments. For ease of reference we have made a table showing how we have addressed each comment, and this has been uploaded along with the manuscript files.

Thank you for arranging peer review and for your consideration of this revised and improved version.

Sincerely,

Mishal Khan